Anterior cruciate ligament tibial avulsion fracture is a common type of anterior cruciate ligament injury. The traditional treatment method is to perform open reduction and internal fixation for repair. However, with the popularization of arthroscopy, arthroscopic high-strength nonabsorbable sutures are also a surgical method chosen by many doctors in recent years. However, there are different opinions on how to choose between the two surgical methods in clinical practice. Therefore, in order to compare the difference between these two operations for the treatment of tibial avulsion fractures of the anterior cruciate ligament, the authors conducted the current study. This provides a basis for clinical practice. They found that both small incision open reduction and internal fixation and arthroscopic high-intensity nonabsorbable sutures can achieve good results in the treatment of tibial avulsion fractures of the anterior cruciate ligament. The operation time of arthroscopic high-intensity nonabsorbable sutures is slightly longer, but the complication rate is lower. Very interesting study. The article is sufficiently novel and very interesting to warrant publication. The Results are presented clearly and have been discussed well.

Scientific Quality: Grade B (Very good)
Language Quality: Grade A (Priority publishing)
Conclusion: Minor revision

Answer: Thank you very much for your recognition, and we will continue to work hard, thank you for your guidance in your busy schedule.